Author's response to reviews

Title: A New Model of Delirium Care in the Acute Geriatric Setting: Geriatric Monitoring Unit

Authors:

Mei Sian Chong (Mei_Sian_Chong@ttsh.com.sg)
Mark Peng Chew Chan (Peng_Chow_Chan@ttsh.com.sg)
Jasmine Kang (Jasmine_Kang@ttsh.com.sg)
Huey Charn Han (Huey_Charn_Han@ttsh.com.sg)
Yew Yoong Ding (Yew_Yoong_Ding@ttsh.com.sg)
Thai Lian Tan (Thai_Lian_Tan@ttsh.com.sg)

Version: 5 Date: 11 August 2011

Author's response to reviews: see over
11\textsuperscript{th} August 2011

TO:
Editor
BioMed Central (BMC Geriatrics)
236 Gray’s Inn Road
London
WC1X8HB

RE: Submission of “A New Model of Delirium Care in the Acute Geriatric Setting: Geriatric Monitoring Unit” study protocol for consideration for publication in BMC Geriatrics

Dear Editor,

We enclose our revised manuscript titled “A New Model of Delirium Care in the Acute Geriatric Setting: Geriatric Monitoring Unit” study protocol for consideration for publication in BMC Geriatrics.

We have made the editorial changes to adhere to journal style as recommended by your editorial production team. There has been no major changes in the content or text changes. The changes include the following:

1. Changes in author list written exactly as in submission system (please note for Chinese, we have two first names (not one first and one middle name)
2. We have added in that “All authors read and approved the final manuscript” in the author’s contributions portion
3. With regards to references 47 and 48, there was incorrect numbering in the last 3 references. We’ve rectified the error in the reference portion
4. The tables have been formatted to the specifications as instructed by the publication team.
5. We have made all the minor revisions as recommended by the publication team.

Please do let us know if we had omitted anything else.

We look forward to hearing from you.

Yours truly,

Dr Chong Mei Sian
Senior Consultant
Department of Geriatric Medicine
Tan Tock Seng Hospital
11 Jalan Tan Tock Seng
S208433. Singapore
Email: Mei_Sian_Chang@ttsh.com.sg
Contact number: (65) 63577859
Fax number: (65) 63524436